Author's response to reviews

Title: The psychometric properties of the 'Hospital Survey on Patient Safety Culture' in Dutch hospitals

Authors:

Marleen Smits (m.smits@nivel.nl)
Ingrid Christiaans-Dingelhoff (ingrid.christiaans@vumc.nl)
Cordula Wagner (c.wagner@nivel.nl)
Gerrit van der Wal (g.vanderwal@vumc.nl)
Peter P Groenewegen (p.groenewegen@nivel.nl)

Version: 4 Date: 25 September 2008

Author's response to reviews:

Statement added to the methods section that formal ethical approval was not needed for this study.